Author's response to reviews

Title: Evaluation of oral microbiology lab curriculum reform

Authors:

Min Nie (niemin@whu.edu.cn)
yan Zhen Gao (973066427@qq.com)
yu Xin Wu (645933142@qq.com)
xi Chen Jiang (354454133@qq.com)
hui Jia Du (137711727@qq.com)

Version: 5  Date: 23 April 2015

Author's response to reviews:

Dear Mr Aldrin Ulep,

We thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Evaluation of oral microbiology lab curriculum reform”. (ID: 1689765945146228)

Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and the main corrections in the paper and the responds to the reviewer's comments are fallowing.

We sought the professional editing service to correct our language, BioMed Central recommends Edanz (http://www.edanzediting.com/bmc1). We have tried our best to revise our manuscript according to the comments, we hope meet with approval.

Revised portion are marked in red in the paper. Attached please find the revised version, including tracked copy and clear copy, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Min Nie
Tel: 15327120298
e-mail: niemin@whu.edu.cn